\batchmode
\documentstyle[12pt]{article}
\makeatletter


\makeatother
\newenvironment{tex2html_wrap}{}{}
\setlength{\textheight}{10in}\begin{document}
\pagestyle{empty}
\newpage

{\samepage \clearpage $$\begin{array}{ll}
 Style   &   Assignment  \\ 

{\rm Narrative}  & {\rm Describe \; a \; personal \; experience \; where   } \\  
                 & {\rm you \; think\;statistical\; knowledge\; would  } \\ 
                 &{\rm have\; been\; helpful\; to\; you.}\\ 
 &\\ 

{\rm Expository}	&{\rm  Describe\; for\; a\; non-specialist\; the\; key  } \\ 
                 &{\rm  ideas\; of obtaining\; information\; about\; a  }  \\ 
                 &{\rm population\; by\; sampling, citing\; examples\;  }\\ 
                 &{\rm in\; current\; news\; stories.}\\ 
 &\\ 
{\rm Critique	}  &{\rm Debunk\; three\; examples\; of\; graphical   }\\ 
           &{\rm abuse\; that \;you\;find \;in\; the\; popular\; press.} \\ 
&\\  
{\rm Analysis	}  &{\rm Simpson's \; paradox \; (pitfalls \; of\; aggregation)  } \\ 
                 &{\rm in \;data on \; race \; and \; imposition \; of \; the  } \\ 
                 &{\rm death \; penalty.}\\ 
&\\  
{\rm Argument	}  &{\rm 	Should\; there\; be\; mandatory\; AIDS\; screening\;}\\  
                 &{\rm  of\; health\; care \; workers?} \\ 
&\\  
{\rm Research \;}&{\rm  Research \;paper \;on \; your \; choice \; of \; CHANCE \;} \\ 
 \;\;\;({\rm final\; project}) &{\rm topic \;in\; the\; news.} 

\end{array}
$$
}


\stepcounter{section}

\end{document}